Author's response to reviews

Title: HNPCC versus sporadic microsatellite-unstable colon cancers follow different routes toward loss of HLA class I expression

Authors:

Jan Willem F Dierssen (JWFDierssen@gmail.com)
Noel FCC de Miranda (N.F.Miranda@lumc.nl)
Soldano Ferrone (Soldano.Ferrone@roswellpark.org)
Marjo van Puijenbroek (M.van_Puijenbroek@lumc.nl)
Cees J Cornelisse (C.J.Cornelisse@lumc.nl)
Gert Jan Fleuren (G.J.Fleuren@lumc.nl)
Tom van Wezel (T.van_Wezel@lumc.nl)
Hans Morreau (J.Morreau@lumc.nl)

Version: 4 Date: 22 February 2007

Author's response to reviews:

Dear editor,

We have made all the formatting changes recommended by your editorial production team.

- We removed all highlighting and tracking
- We now spell Authors' contributions
- The figures legends are place before the tables
- We removed the green arrows from table 2. They were error messages from excel worksheet.
- We converted Table, 3, 4 and 5 into Figures so that the graphical representation is printable. We have corrected all the numbering of previous figures and altered the citations throughout the text. Please note that the figures submitted previously will be again submitted with the proper numbering.
- We have titled the figures section with "Figure legends"

Thank you for your detailed revision of the manuscript

With best regards,

Noel de Miranda